Author’s response to reviews

Title: Head CT is of limited diagnostic value in critically ill patients who remain unresponsive after discontinuation of sedation

Authors:

Jay S Balachandran (jbalachandran@partners.org)
Mairaj Jaleel (MJaleel@uwhealth.org)
Manu Jain (m-jain@northwestern.edu)
Niraj Mahajan (nirajmahajan@hotmail.com)
Ravi Kalhan (r-kalhan@northwestern.edu)
Rajesh Balagani (rbalagani@gmail.com)
Helen K Donnelly (h-donnelly@northwestern.edu)
Eugene Greenstein (egreenst@md.northwestern.edu)
Gokhan M Mutlu (g-mutlu@northwestern.edu)

Version: 8 Date: 29 April 2009

Author’s response to reviews: see over
April 29, 2009

Melissa Norton, MD
Editor-in-Chief
BMC Journals

Re: MS: 6202835792131180 - Head CT is of limited diagnostic value in critically ill patients who remain unresponsive after discontinuation of sedation

Dear Dr. Norton:

We are excited that our manuscript was accepted for publication in the BMC Anesthesiology journal. Below please find point-by-point response to the revisions required by the journal. All the authors reviewed and approved the final version of the manuscript. None of the authors has any conflict of interest related to the findings in this manuscript.

Thank you.

Sincerely,

Gokhan M. Mutlu, MD
Associate Professor of Medicine
Northwestern University Feinberg School of Medicine
Major revisions

C1. “Tables - Tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.”

R1. We recreated the tables using the table tool in MS Office Word.